Author’s response to reviews

Title: Persistent Unexplained Chest Pain and Dyspnea in a Patient With Coronary Artery Disease: A Case Report

Authors:

Sameh Mobarek (skmobarek@gmail.com)

Version: 2 Date: 16 Jan 2020

Author’s response to reviews:

January 16, 2020
Sarah Gray
Editor
BMC Cardiovascular Disorders

Dear Ms Gray,
Thank you for considering a revised version of my manuscript titled “Persistent Unexplained Chest Pain and Dyspnea in a Patient With Coronary Artery Disease: A Case Report” (Manuscript No: BCAR-D-19-00272R1) for publication in BMC Cardiovascular Disorders. I have addressed the revisions requested below.

I look forward to hearing your final decision on my manuscript soon.

Best regards,

Sameh K. Mobarek, MD, FACC
North Carolina Heart and Vascular Specialists
UNC/Rex Healthcare
1505 SW Cary Parkway
Suite 300
Cary, NC 27511
Email: skmobarek@gmail.com
Phone: (919) 387-3260

Assistant Editor Comments

1. List of Abbreviations
We note that some abbreviations used in the manuscript text have not been included in the ‘List of Abbreviations’ section. Please ensure that all abbreviations present in the text are listed in this section. For example nOH, BP and HIV.
Response: All abbreviations have now been spelled out in the manuscript.

2. Acknowledgements
Please remove the funding information from the Acknowledgements and include it in the Funding
section instead.
Response: This change has been made (page 9).

3. **Title Page**
Please remove the declaration of interest, sources of funding and role of the funding source from the title page and ensure this information is include in the appropriate sections of the Declarations.
Response: This information has been removed from the title page and appears in the appropriate sections on pages 8–10.

4. **Remove files**
Duplicate manuscript, care checklist, r2rs and anything do not wish to be published
Please remove the response to the reviewers comments, the CARE Checklist, the duplicate manuscript and any other files you do not wish to be published from the file inventory as they are no longer needed at this stage of the editorial process.
Response: All files that are no longer needed have been removed.

5. **Clean manuscript**
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: A clean version has been uploaded as requested. The figure included with the submission is cited in the manuscript.

**Reviewer 2 Comments**
1. The authors have addressed adequately all my comments, including adding those details which were not present in the first version, and acknowledging the limitations inherent to the work.
Response: Thank you for your review.